Author’s response to reviews

Title: Relapsed Papillary Urothelial Neoplasm of Low Malignant Potential (PUNLMP) of the young age: a case report and a review of the literature

Authors:

Palma Maurizi (palma.maurizi@unicatt.it)
Michele Antonio Capozza (micheleantoniocapozza@gmail.com)
Silvia Triarico (silviatriarico@libero.it)
Maria Luisa Perrotta (mperrrotta@scamilloforlanini.rm.it)
Vito Briganti (vito.briganti@fastwebnet.it)
Antonio Ruggiero (antonio.ruggiero@unicatt.it)

Version: 2 Date: 28 Apr 2019

Author’s response to reviews:

Editor Comments:

1. Ethics approval and consent to participate

We have amend our statement to read ‘Not applicable’.

2. Authors’ contributions

we have better described the role of each author, ensuring that all listed authors are reflective of ICMJE guidelines.